Author’s response to reviews

Title: Descending Necrotizing Mediastinitis Caused by Streptococcus Constellatus in an Immunocompetent Patient: Case Report and Review of the Literature

Authors:

rui-hai ye (yehoo2003@163.com)
jun chao yang (yangjunchaozj@163.com)
hui hua hong (honghuiha@sina.com)
yu ping mao (2242723113@QQ.com)
yuan hong zhu (zhuyuan0487@126.com)
yu cao (caoyumail@163.com)
zhen wang (wangzhen610@sina.cn)

Version: 2 Date: 15 Aug 2019

Author’s response to reviews:

Dear editor:

I'm sorry that we can't attach the detailed information, so it caused your misunderstanding. I hope the supplementary information can make up for our previous shortcomings. If there are any questions about it, please contact us. (Email yehoo2003@163.com). Thank you!

Yehoo

2019.8.15